Author's response to reviews

Title: Cystic lymphangioma of broad ligament: a case report.

Authors:

   K Harish Dr (drkhari@yahoo.com)
   S R Karthik Dr (drkarthiksrishi@gmail.com)
   C S Manjunath Dr (manzunath@hotmail.com)

Version: 6 Date: 22 May 2008

Author's response to reviews: see over
To
The Editor
BMC Journal of medical case reports
Sir/ Madam

Subject: Submission of Case Report after peer review

MS: 2262556571787266

Thank you.
The reviewers’ comments are as under:

Reviewer 1
Erkan Yildirim
Suggested alterations: None

Reviewer 2
Oluwole Fadare

Suggested alterations: Specify the nature of the cells lining the cysts. e.g low cuboidal, flat/attenuated etc.

Authors’ response: In the description of the photomicrograph, the description includes “smooth muscle in the wall of cyst indicated by white bold arrow and flattened endothelial like cells lining the cyst wall indicated by bold black arrow”

The article is hence resubmitted without any alteration

Thanking you

Sincerely yours

Dr K Harish